Author's response to reviews

Title: Ambient biomass smoke and cardio-respiratory hospital admissions in Darwin, Australia

Authors:

Fay H Johnston (fay.johnston@utas.edu.au)
Ross S Bailie (ross.bailie@menzies.edu.au)
Louis S Pilotto (l.pilotto@unsw.edu.au)
Ivan C Hanigan (ivan.hanigan@anu.edu.au)

Version: 5 Date: 7 September 2007

Author's response to reviews:

To the editors BMC Public health

Please revise as follows:
1. Page 4 line 2 please move the position of the full-stop from before citation 8 to after it.
2. Page 5 Methods section, study design paragraph. In the first line include a hyphen to correct 'case crossover' to 'case-crossover'
3. Page 5 Methods section, study design paragraph. Reword the second line to change 'the days they were admitted' to 'the day each case was admitted'

Many thanks for your help
Dr Fay Johnston